Author’s response to reviews

Title: The acute effects of alcohol on auditory thresholds

Authors:

Tahwinder Upile (mrtupile@yahoo.com)
Fabian Sipaul (mr2fabporsche@hotmail.com)
Waseem Jerjes (waseem_wk1@yahoo.co.uk)
Sandeep Singh (sandeepupile@yahoo.com)
Reza Nouraei (reza.nouraei@gmail.com)
Mohammed El Maaytah (elmaaytah@hotmail.com)
Peter Andrews (peter.andrews@royalfree.nhs.uk)
John Graham (john.graham@royalfree.nhs.uk)
Colin Hopper (c.hopper@ucl.ac.uk)
Anthony Wright (anthony.wright@ucl.ac.uk)

Version: 5 Date: 13 September 2007

Author’s response to reviews:

Dear Sir/Madame,

All the requested changes took place.

Many thanks.

The authors